Author’s response to reviews

Title: Therapeutic angiogenesis in cardiovascular disease

Authors:

Hilal Al Sabti (alsabti@hotmail.com)

Version: 4 Date: 10 November 2007

Author’s response to reviews:

Dear Sir,,, I am submitting the change of Formate requested from me as follow:
1) References: I changed the References as there are now cited using consecutive numbers in bracket. I also omitted all the superscript numbers. All the references are now placed before the full stop.

2) Authors’ contributions are now added and this is a review article done only by one Author and approved by the same author.

3) all section heading have been changed to sentence case and the unnecessary capitals were removed.

4) The references do not contain et al term and the names of the authors available.

THank you very much.